Author’s response to reviews

**Title:** Kinesin Eg5 inhibition by 3,4-dihydropyrimidin-2(1H)-one (or thione) derivatives impairs multiple tumorigenic properties inducing breast cancer cells to apoptosis

**Authors:**

Bruna C Guido (brunacguido@gmail.com)
Luciana M Ramos (lucianamramos@hotmail.com)
Diego O Nolasco (nolasco@mit.edu)
Catharine C Nobrega (catharine_carrara@hotmail.com)
Bárbara YG Andrade (barbarayasm@ gmail.com)
Aline Pic-Taylor (alinepic@unb.br)
Brenno AD Neto (brenno.ipi@gmail.com)
Jose R Correa (correa@unb.br)

**Version:** 3  
**Date:** 21 November 2014

**Author’s response to reviews:**

As suggested by Ms. Cherry Battad, line numbers were included in the main manuscript, but no other changes were made.